Author's response to reviews

Title: Trends in completion of immunization in the context of health reforms: The Tanzania case study

Authors:

Innocent A Semali IAS (isemali@muhas.ac.tz)

Version: 2 Date: 1 October 2009

Author's response to reviews: see over
Dear Sirs,

We are kindly submitting our paper to you to kindly consider it for publication in the Health System Research Journal. The paper is titled: **Trends in completion of immunization in the context of health reforms: The Tanzania case study.** It is a paper authored by Innocent A Semali.

It was perceived that there was a need to explain the observed stagnation in completion of immunization in Tanzania. The author accessed the Tanzania Demographic and Health Survey for 1996 and 2004 through the internet after a formal request submitted. Data on immunization and explanatory variables was extracted from the large data sets. Analysis was done using STATA 9.0 and the results are presented in this paper.

The author analysed the data as well as writing the manuscript. The author also declare no conflict of interest.

The corresponding author’s particulars are as follows:

**Name:** Innocent A Semali  
**Address:** School of Public Health and Social Sciences  
Muhimbili University of Health and Allied Sciences  
P.O. Box 65015  
Dar es Salaam  
Tanzania.
Telephone: +255(0)754269838 ; email: isemali@muhas.ac.tz

With kind regards.

I A Semali